Specific Comments to Authors: The purpose of this article is to evaluate the safety and efficacy of enhanced recovery after surgery in inflammatory bowel disease surgery. In conclusion, enhanced recovery after surgery was effective for the quicker recovery in inflammatory bowel disease surgery and did not lead to increased complications. The title correctly reflects the main subject of the manuscript. The abstract summarizes and reflects the work described. The key words reflect the focus of the manuscript. The manuscript adequately describes the background, present status and significance of the study. The findings are stated in a clear and definite manner. The discussions are accurate and clear. The manuscript is well, concisely and coherently organized and presented. The style, language and grammar is accurate and appropriate. The manuscript meets all the requirements for a work of this type. My assessment is that the overall quality of this article is excellent and should be accepted with high priority.

Answer: Thanks for your carefully reading. It’s our honor to receive your comments, and we are appreciated with your approval of our article.

Company editor-in-chief:
I have reviewed the Peer-Review Report, the full text of the manuscript, and the relevant ethics documents, all of which have met the basic publishing requirements of the World Journal of Clinical Cases, and the manuscript is conditionally accepted. I have sent the manuscript to the author(s) for its revision according to the Peer-Review Report, Editorial Office’s comments and the Criteria for Manuscript Revision by Authors. The title of the manuscript is too long and must be shortened to meet the requirement of the journal (Title: The title should be no more than 18 words). Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor. Authors are required to provide standard three-line tables, that is, only the top line, bottom line, and column line are displayed, while other table lines are hidden. The contents of each cell in the table should conform to the editing specifications, and the lines of each row or column of the table should be aligned. Do not use carriage returns or spaces to replace lines or vertical lines and do not segment cell content.

Answer: Thanks for your reminder. The title was changed into “The effect of enhanced recovery after surgery on inflammatory bowel disease surgery: A meta-analysis”. The figures were uploaded using PowerPoint. Table 1 and Table 2 were made by standard three-line tables.